Author's response to reviews

Title: Immune responses in patients with HIV infection after vaccination with recombinant Hepatitis B virus vaccine

Authors:

Neelam Pasricha (pgimer@chd.nic.in)
Usha Datta (pgimer@chd.nic.in)
Yogesh Chawla (pgimer@chd.nic.in)
Surjit Singh (pgimer@chd.nic.in)
Sunil K Arora (pgimer@chd.nic.in)
Archana Sud (pgimer@chd.nic.in)
Ranjana W Minz (pgimer@chd.nic.in)
Biman Saikia (bimansaikia@hotmail.com)
Haqeeqat Singh (pgimer@chd.nic.in)
Isaac James (pgimer@chd.nic.in)
Shobha Sehgal (pgimer@chd.nic.in)

Version: 5 Date: 24 March 2006

Author's response to reviews:

The figures have been cropped with the help of the online guide.
The Bibliography section has been changed to "References"
Acknowledgement section has been added along with the source of funding.